Author’s response to reviews

Title: Durability assessment results suggest a serviceable life of two, rather than three, years for the current long-lasting insecticidal (mosquito) net (LLIN) intervention in Benin

Authors:

Virgile Gnanguenon (amerusangel@yahoo.fr)
Roseric Azondekon (roseric_2000@yahoo.fr)
Frederic Oke-Agbo (amerusangel@yahoo.fr)
Raymond Beach (rfb@yahoo.fr)
Martin Akogbeto (akogbeto@yahoo.fr)

Version: 2 Date: 12 November 2013

Author’s response to reviews:

Dear editor,

Many thanks for your comments. Following your request, the changes below has been brought to the manuscript:

- The text of Figure 1 legend (in figure legend section, page 15) has been amended to make clear that it is a reproduced copy with reference to Parasites & Vectors manuscript (reference 11). I also confirmed that we own a copyright of the map (Figure 1. Reproduced copy (with copyright) of Benin Map LLIN tracking assessment sites: Kessounou, Allada, Kandi and Malanville [11] ).

Regards,

Virgile GNANGUENON